Author's response to reviews

Title: Subdural spread of injected local anesthetic in a selective transforaminal cervical nerve root block: a case report

Authors:

Katsuhiro Tofuku (tofuku@m2.kufm.kagoshima-u.ac.jp)
Hiroaki Koga (gpmhiroaki@yahoo.co.jp)
Setsuro Komiya (skomiya@m3.kufm.kagoshima-u.ac.jp)

Version: 2 Date: 8 November 2011

Author's response to reviews: see over
November 1, 2011

Professor Michael Kidd AM
Editor-in-Chief, Journal of Medical Case Reports
Flinders University
Australia

Dear Dr. Kidd:

We would like to submit our manuscript entitled “Subdural spread of injected local anesthetic in a selective transforaminal cervical nerve root block” for consideration for publication in Journal of Medical Case Reports. We are submitting our manuscript as a Case Report. We believe that this is a rare case of high spinal cord anesthesia due to inadvertent subdural spread of injected local anesthetic in a selective transforaminal cervical nerve root block and therefore will interest the readers. Our manuscript provides important information about risk potential of serious complications in selective cervical nerve root block using transforaminal approach, which will be of interest to clinicians. This manuscript has not been published nor is it currently under consideration for publication by any other journal. We will be grateful if this manuscript is considered for publication in Journal of Medical Case Reports. Thank you very much.

Sincerely,

Katsuhiro Tofuku, M.D., Ph.D.
Department of Orthopaedic Surgery
Imakiire General Hospital
4-16 Shimotatsuo, Kagoshima 892-8502, Japan
Tel: +81-99-226-2211
Fax: +81-99-222-7906
E-mail: tofuku@m2.kufm.kagoshima-u.ac.jp